Author's response to reviews

Title: Active surveillance of Q fever in human and animal population of Cyprus

Authors:

- Fidias Loukaides (annapsa@med.uoc.gr)
- Christos Hadjichristodoulou (xhatzi@med.uth.gr)
- Elpidoforos S. Soteriades (esoteria@hsph.harvard.edu)
- Virginia Kolonia (xhatzi@med.uth.gr)
- Maria-Christina Ioannidou (annapsa@med.uoc.gr)
- Anna Psaroulaki (annapsa@med.uoc.gr)
- Yannis Tselentis (annapsa@med.uoc.gr)

Version: 5 Date: 7 March 2006

Author's response to reviews: see over
Larissa, March 7, 2006

To: The BioMed Central Editorial Team

Re: “ACTIVE SURVEILLANCE OF Q FEVER IN HUMAN AND ANIMAL POPULATION OF CYPRUS”

Dear Editor

We are pleased to submit a revised manuscript for your consideration. We have made every effort to make all the formatting changes you suggested.

Sincerely,

Christos Hadjichristodoulou, MD, PhD
Assistant Professor
Department of Hygiene and Epidemiology
22 Papakyriazi Street
41222 Larisa
Greece
Tel: +30 2410 – 565259
Fax: +30 2410 – 565259
E-mail: xhatzi@med.uth.gr